Author's response to reviews

Title: Acral Post Traumatic Tumoral Calcinosis in Pregnancy: A Case Report

Authors:

Nick Hutt (nickhutt2@hotmail.com)
Paul Baghla (paulbaghla@blueyonder.co.uk)
Vivek Gulati (vivek0@hotmail.com)
Philip Pastides (ppastides@hotmail.com)
Mike Beverly (mikebeverly@hotmail.com)
Wassim Bashir (wassimbashir24@hotmail.com)

Version: 4 Date: 20 December 2010

Author's response to reviews:

Many thanks for your feedback. The following changes have been preformed:

1) The competing interest and consent sections have been split
2) A brief conclusion has been added to the end of the manuscript